Author’s response to reviews

Title: A Simulation Study of Sample Size for Multilevel Logistic Regression Models

Authors:

Rahim Moineddin (rahim.moineddin@utoronto.ca)
Flora I Matheson (mathesonf@smh.toronto.on.ca)
Richard H Glazier (rick.glazier@ices.on.ca)

Version: 5 Date: 18 June 2007

Author's response to reviews: see over
Dear Iraxe Puebla:

We thank the reviewer and yourself for the final comments on our paper and have made the following changes as requested:

(a) The findings from the additional simulation studies have been summarized in table format (please see page 13 and Table 5 on page 23).
(b) Initials are used when referring to author contributions (please see page 15).
(c) Cited references within the text now are formatted with consecutive numbers in brackets.
(d) “Biomedcentral” style for Reference Manager was used to format the citations and references.
(e) The manuscript has been reviewed to ensure that our formatting conforms to the manuscript formatting checklist.
(f) Minor editorial changes have been completed.

Thank you for the conditional acceptance of this manuscript. We hope that our paper now meets with your approval as we welcome the opportunity to publish in BMC Medical Research Methodology.

I look forward to hearing from you.

________________________________________
Rahim Moineddin, Ph.D.
Department of Family and Community Medicine
University of Toronto
Toronto, Ontario
Rahim.moineddin@utoronto.ca
(416) 946-5860